Author’s response to reviews

Title: Methamphetamine use and correlates in two villages of the highland ethnic Karen minority in northern Thailand: a cross sectional study

Authors:

Eiko Kobori (kobori.e@kw8.ecs.kyoto-u.ac.jp)
Surasing Visrutaratna (surasing@loxinfo.co.th)
Yuko Maeda (maeda@hs.med.kyoto-u.ac.jp)
Siriporn Wongchai (siriporn_cm@hotmail.com)
Akiko Kada (kada@hsp.ncvc.go.jp)
Masako Ono-Kihara (okmasako@pbh.med.kyoto-u.ac.jp)
Yoko Hayami (yhayami@cseas.kyoto-u.ac.jp)
Masahiro Kihara (poghse@pbh.med.kyoto-u.ac.jp)

Version: 4 Date: 1 May 2009

Author’s response to reviews:

Dear Sir
Editorial Office
BMC International Health and Human Rights

I would like to submit the latest manuscript, titled “Methamphetamine use and correlates in two villages of the highland ethnic Karen minority in northern Thailand: a cross sectional study,” to BMC International Health and Human Rights.

Belows are points corrections were made this time.

1. Email address of Eiko KOBORI was changed to "kobori.e@kw8.ecs.kyoto-u.ac.jp" both in the manuscript and the submission system.
2. Page numbers typed at footer were deleted.
3. "Urine test," typed in bold font at the 5th line from the bottom in the 1st paragraph in Background section was changed to Times New Roman.
4. "=," typed in the 2nd line in the 1st paragraph in Results section was deleted.
5. "n" was added at the word of "Karen," which originally typed as "Kare" at the 4th line from the bottom; and the sentence "All authors read and approved the final manuscript" was added in the last line in the section of Authors’ contributions.
6. All of 3 tables were revised as advised by the Editorial Production Team.

7. The last page of the manuscript (confirmation that the English has been checked) was deleted.

8. The 6th author's name was revised both in the manuscript and the submission system.

These are all revisions made this time.

Please kindly let me know if further revisions are necessary.

With best regards,

Eiko KOBORI